Author's response to reviews

Title: Evaluation and optimization of commercial Enzyme Linked Immunosorbent Assay kit for detection of Chlamydophila pneumoniae IgA antibodies

Authors:

Olfa Frikha-Gargouri (Olfafrikha2005@yahoo.fr)
Radhouane GDOURA (gdourar@yahoo.com)
Abir ZNAZEN (abirznazen2001@yahoo.fr)
Nozha BEN ARAB (nozhabenarab@yahoo.fr)
Jalel GARGOURI (jalelgargouri@yahoo.fr)
Mounir BEN JEMAA (mounir.benjemaa@rns.tn)
Adnene Hammami (adnene.hammami@rns.tn)

Version: 3 Date: 29 June 2008

Author's response to reviews:

Dear Editor,

We are very grateful to you for your response to our manuscript entitled "Evaluation and optimization of commercial Enzyme Linked Immunosorbent Assay kit for detection of Chlamydophila pneumoniae IgA antibodies" MS: 1140674523185246

Would you please find enclosed the revised form of our manuscript.

I- Responses to editor’s notes

1. We recommend that you copyedit the paper to improve the style of written English, particularly the Abstract. If this is not possible, you may need to use a professional copyediting service. Examples are those provided by the Manuscript Presentation Service (www.biomedes.co.uk), International Science Editing (http://www.internationalscienceediting.com/) and English Manager Science Editing (http://www.sciencemanager.com/). BioMed Central has no first-hand experience of these companies and can take no responsibility for the quality of their service.

We have used the English Manager Science Editing service to improve the quality of the entire article. Almost all sentences have been changed but with keeping the same meaning.

2. Please state in the Methods whether informed consent given by subjects was written or verbal.

Informed consent given by subjects was verbal and was added in lane 87 in page 5.

Please also ensure that your revised manuscript conforms to the journal style
It is important that your files are correctly formatted.

Our revised manuscript was also formatted according to the Journal style.

We will be ready for any suggestion or further modification that you need me to make.

Looking forward to having our manuscript published in your Journal, we would like to send you my highest respect.

Sincerely yours

Radhouane GDOURA